Author’s response to reviews

Title: A case report of iatrogenic deterioration of yet undiagnosed Rhombencephalitis; always be careful with corticoids.

Authors:

Loes Mandigers (l.mandigers@erasmusmc.nl)
Jelle Epker (j.epker@erasmusmc.nl)

Version: 3 Date: 22 Nov 2018

Author’s response to reviews:

Rotterdam, November 22, 2018

Dear Editor,

We are delighted to hear that our case report is potentially acceptable for publication in your journal, BMC Clinical Pathology. As requested, we made the following changes to the manuscript, as described in a point-by-point reply, listed below.

Yours sincerely,

Loes Mandigers and Jelle Epker

1. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

   We included a heading “Figure Legends” on page 8 consisting of all Figure titles.
2. Figure files should contain only the image/graphic, as well as any associated keys/annotations. Where titles/legends are present within the figure files, please remove them.

All titles in the figure files were removed.

3. Thank you for providing your consent form. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.

We removed the consent form from our manuscript.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The manuscript and the accompanying figures are now uploaded as a clean versions without tracked changes or comments.

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English.

The complete manuscript is checked for English language by someone who speaks English fluently.